Author’s response to reviews

Title: Factors that influence patient empowerment in inpatient chronic care: Early thoughts on a diabetes care intervention in South Africa

Authors:

Nina Thandi Abrahams (ntabrahams@gmail.com)

Lucy Gilson (lucy.gilson@uct.ac.za)

Naomi Levitt (naomi.levitt@uct.ac.za)

Joel Dave (joel.dave@uct.ac.za)

Version: 2 Date: 22 Nov 2019

Author’s response to reviews:

Comment: In the "Consent for publication" section, please clarify if written informed consent for publication was obtained for the use of all quotes.

Response: Sentence added on page 26 under "Consent for publication" section. "Written informed consent as well as verbal clarification regarding publishing participant quotes was obtained from all participants before interviews took place."

A clean, final version of the manuscript has been uploaded.